Author's response to reviews

Title: Limitations in a frataxin knockdown cell model for Friedreich ataxia in a high-throughput drug screen

Authors:

Nadège Calmels (nadege.calmels@chru-strasbourg.fr)
Hervé Seznec (seznech@cenbg.in2p3.fr)
Pascal Villa (pascal.villa@pharma.u-strasbg.fr)
Laurence Reutenauer (Laurence.Reutenauer@titus.u-strasbg.fr)
Marcel Hibert (mhibert@aspirine.u-strasbg.fr)
Jacques Haiech (haiech@aspirine.u-strasbg.fr)
Pierre Rustin (rustin@rdebre.inserm.fr)
Michel Koenig (mkoenig@titus.u-strasbg.fr)
Hélène Puccio (hpuccio@igbmc.u-strasbg.fr)

Version: 4 Date: 20 August 2009

Author’s response to reviews:

Dear Editor,

We are delighted that you have accepted our manuscript. We have made all of the formatting requirement asked:

1. Title- identical as submission system
2. Author list
3. References: updates ref 21, 47, 50
4. Made figure citation consistent throughout the text
5. Added email of each author on title page
6. Changed underlining by bold
7. Cropped as much as possible the figures

Please let me know if there are any other changes necessary.

 Regards
Hélène Puccio